Author's response to reviews


Authors:

Jesus E Duenas-Arias (jedanet@yahoo.com)
Maribel Aguilar-Medina (maribela2@excite.com)
Elikym Arambula-Meraz (eliakymarambula@hotmail.com)
Juliana B Valenzuela-Camacho (juliana_70@yahoo.com)
Angelina Vega-Solano (angyqfb@hotmail.com)
Julio Granados (julgrate@yahoo.com)
Rosalio Ramos-Payan (ramospayan@yahoo.com.mx)

Version: 6 Date: 14 September 2007

Author's response to reviews: see over
Enclosed please find the manuscript (above) with the format corrections.

Author’s names in article are now the same than in the submission system.
Title page was kept on one page.
Heading and subheadings have the correct format.
Keywords were removed.
References were renumbered.
Figures were deleted from the text and uploaded in submission system.

I hope you will find it now suitable for publication.

Sincerely yours,

Rosalío Ramos Payán
FCQB-UAS